University of Chicago Law School Chicago Unbound

Coase-Sandor Working Paper Series in Law and Economics

Coase-Sandor Institute for Law and Economics

2005

Paretian Intergenerational Discounting

David A. Weisbach

Dexter Samida

Follow this and additional works at: https://chicagounbound.uchicago.edu/law_and_economics Part of the Law Commons

Recommended Citation

David A. Weisbach & Dexter Samida, "Paretian Intergenerational Discounting" (John M. Olin Program in Law and Economics Working Paper No. 255, 2005).

This Working Paper is brought to you for free and open access by the Coase-Sandor Institute for Law and Economics at Chicago Unbound. It has been accepted for inclusion in Coase-Sandor Working Paper Series in Law and Economics by an authorized administrator of Chicago Unbound. For more information, please contact unbound@law.uchicago.edu.

CHICAGO

JOHN M. OLIN LAW & ECONOMICS WORKING PAPER NO. 255 (2D SERIES)



Paretian Intergenerational Discounting

Dexter Samida and David A. Weisbach

THE LAW SCHOOL THE UNIVERSITY OF CHICAGO

August 2005

This paper can be downloaded without charge at: The Chicago Working Paper Series Index: http://www.law.uchicago.edu/Lawecon/index.html and at the Social Science Research Network Electronic Paper Collection: http://ssrn.com/abstract_id=786546

Paretian Intergenerational Discounting

Dexter Samida and David A. Weisbach

The University of Chicago Law School

August 15, 2005

<u>Abstract</u>

This paper argues that discounting costs and benefits of projects for the opportunity costs of capital Pareto dominates decision criteria that do not discount. It considers and rejects several objections to the Pareto dominance argument, including the problem of making compensating transfers for the costs and benefits of projects and whether taking opportunity costs into account is different than discounting. It also argues that discounting future costs and benefits of projects does not under-value future generations.

Preliminary Draft. Please do not cite without permission. Send comments to: <u>d-weisbach@uchicago.edu</u>

Paretian Intergenerational Discounting

Dexter Samida^{*} and David Weisbach^{**} August 15, 2005

The problem presented by intergenerational discounting is well known. Suppose we are going to engage in a project that creates costs and benefits that are separated over a long period of time. The most common examples are global warming and disposal of nuclear wastes, but the problem extends to any project that affects more than one period, future or past. How are we to compare the costs and benefits? In private projects, costs and benefits are always discounted because of the opportunity cost of capital. But if we discount public projects over long periods of time, future lives can seem to be significantly undervalued compared to present lives. With a sufficiently high discount rate or a sufficiently long time period, future lives will seem to have a value vastly lower than present lives.

Much of the discussion of discounting is driven by the intuitive implausibility of such comparisons. The power of discounting over long periods can astonish even those deeply familiar with the concept.¹ It is easy to construct examples where saving a single life today means allowing most of the world to perish at some date in the future. In one of our favorite quips, two commentators ask you to "[i]magine finding out

^{*}Clerk, Judge Frank Easterbrook, United States Court of Appeals for the Seventh Circuit

^{**}Walter J. Blum Professor, The University of Chicago Law School. We thank Eric Posner, Adrian Vermuele Cass Sunstein, and participants at a workshop at the University of Chicago for many helpful comments and conversations.

¹Consider the following example, [cite]. Suppose we have a chess board and begin by putting a single penny on the first square. In each subsequent square, we double what was in the previous square, so that the second square has two cents, the third, four cents, and so on. It would be impossible to finish this game because the last square would have to have more wealth than exists in the world today. More realistic but less dramatic examples show that with modest discount rates and time horizons, say 5% and 200 years, saving a billions of in the future is worth sacrificing only thousands of lives today.

Samida	and	Weisbach

that you, having just reached your twenty-first birthday, must soon die of cancer because one evening Cleopatra wanted an extra helping of dessert." Cowen and Parfit [1992; 145] The implication is that no moral theory can support allowing your death to satisfy Cleopatra's sweet tooth. Commentators frequently compare a global catastrophe in the distant future with trivial costs today. Ackerman and Heinzerling [2002; 1571] compare the death of a billion people in 500 years to the death of a single person today. Heal [1998; 13] compares the world GNP in two hundred years to the price of a used car today and, based on the implausibility of this comparison, develops book-length formalization of the theory of sustainable development.² These types of comparisons create unease with discounting even among economists. [Portney and Weyant 1999; 5]

The intuitions driven by these examples and, indeed, even the very mathematics of these examples, are incorrect. We argue here that discounting is appropriate, and, in fact, required by any moral theory that accepts the Pareto principle. In particular, under plausible conditions, discounted cost-benefit analysis Pareto dominants any other decision procedure. Although this argument is well known, it does not seem to have been fully appreciated or has been subject to incorrect criticisms. Substantial movements in environmental law, such as the sustainable development movement, argue for decision criteria that are Pareto dominated, and economists, lawyers, and philosophers frequently argue for a zero discount rate even while seeming to accept the Pareto principle.³

Although Pareto arguments do not necessarily need it, we place the argument within the context of a welfarist, consequentialist ethic, and discuss how it relates to the optimal taxation arguments derived from that ethic. In particular, optimal tax theory can be used to show that under reasonable conditions, cost benefit analysis should not be weighted to

²For a similar formal theory based on the intuition that discounting is inappropriate, see Chichilnisky [1996].

³See., e.g, Parfit [1984]; Cowen [1992]; Revesz [1999].

consider distributional issues, regardless of how egalitarian the decision maker. We discuss the conditions for this theorem and whether they are met in the intergenerational context.

The Paretian arguments made here relate only to opportunity costs or the productivity of capital. Therefore, we do not consider pure time preference arguments for discounting, such as impatience, uncertainty, and the like. Many arguments against discounting focus on whether it is ethical to consider pure time preference. For example, see Revesz [1999; __] We express no opinion on these arguments here. Moreover, because the arguments are based on the Pareto criteria, we need not consider explicitly whether future generations are likely to be better off or worse off than is the current generation, although the optimal taxation framework can incorporate these considerations.⁴

Section I lays out the basic problem and framework for addressing it. Because the arguments about discounting are about justice between generations, we take generations as the basic unit for measuring Pareto improvements rather than arguing about the welfare of each and every individual within each generation across all of time. Section I discusses this approach. Section II considers the argument that discounted cost benefit analysis Pareto dominates any other decision criteria and discusses how this argument relates to the optimal taxation arguments. Section III considers the major objections to the Pareto argument. In particular, Section III considers the problem of the difficulty of making intergenerational transfers, such as when there are intervening generations and considers whether considerations of opportunity costs are distinct from discounting. Section IV briefly considers additional issues, including the use of rhetoric in discounting arguments.

⁴The opportunity cost argument implicitly includes considerations of relative wealth because if capital is productive, giving rise to opportunity costs, future generations are likely to be wealthier than we are.

I. The Problem and the Ethical Framework

Suppose there is a potential project that has costs and benefits dispersed over a long period of time. One of the most important examples is abatement of global warming. Reducing global warming requires the current generation to incur substantial costs while producing benefits hundreds of years into the future. Other prominent examples include nuclear waste disposal and extinction of species. The problem, however, is general and runs in both directions (i.e., costs today, benefits in the future or benefits today, costs in the future). For example, rather than saving resources today to help the future, we could consider the inverse, a project that uses resources today to help current lives when those resources might otherwise have been used in the future. Using up the world's petroleum supplies is such a case, as is not engaging in the global warming abatement. By merely switching the assumed status quo, we can imagine any project running in either direction. By "project" we mean any decision that imposes costs and benefits, including actual projects such as building things, regulatory projects that impose costs and benefits over time, research into new technologies, taxing, saving, running deficits or surpluses, or any other decision that has real world effects.

It is important to note that projects (and decisions about discounting) can involve the past as well as the future. A good example is the measurement of reparations for harms that occurred long ago. We might in the case of reparations for slavery, compensate harms that as long as three or four hundred years ago. Discounting can have an enormous impact on the size of reparations claim.

The project will either pass or fail discounted cost benefit analysis. By this we mean that the marginal costs are either less than or exceed the marginal benefits when all costs and all benefits are discounted (by an appropriate discount rate, discussed later) to the same time period. Note that because projects can involve either preserving or destroying resources, if one is a committed environmentalist, one cannot say whether passing or failing discounted cost benefit analysis helps or hurts the environment. If the project is reparations, discounting increases the transfers to the harmed group and failing to discount reduces the transfers.⁵ Moreover, there is no difference, so far, between discounting to the present period and future valuing to some future period if the same discount rate is used. All that is required is that the marginal costs and benefits be compared at the same point in time. (A discussion of the possible differences between discounting and taking opportunity costs into account can be found in Section III below.)

We also assume that all of the project's costs and benefits are properly counted. There are many discussions in the literature about the inadequacies of various discounting or cost-benefit procedures used in practice. Adler and Posner [2000]; Heinzerling, [1999; 69-71]; Ackerman and Heinzerling [2002; 1563-1570.] While these issues are important, they are not central to the underlying ethical issue of whether discounting is appropriate. We also assume that the value of life (and non-life harms) is properly measured, and that life, risks to life, and enjoyment of life can be measured in money. Posner and Sunstein [2005] This means, among other things, that if the value of life in the future is higher than it is today (say because of increased wealth), this higher value is used. Whether life or enjoyment can be measured with money is an interesting problem but it is orthogonal to the issues presented by discounting because it arises for projects that have immediate effects.⁶

⁵For example, suppose that the harm to each slave was \$1,000 at the time and the harms occurred 250 years ago. With no discounting, reparations would be \$1,000 per harm. With discounting, at a 5% rate, reparations would be almost \$200 million per harm.

⁶Many counter-arguments to discounting rely on an intuition that the value of life cannot be measured. They compare spending money to the certain loss of life. See., e.g., Cowen and Parfit [1992] These examples trigger intuitions not only by the exponential effect of discounting but also by the difference between willingness to pay and willingness to accept. This confounding of intuitions regarding separate issues can create erroneous conclusions regarding either one of the issues.

Samida	and	Weishach

More generally, we will not discuss here whether cost benefit analysis is appropriate for projects that have only current costs and benefits.⁷ In trying to understand objections to discounting, we think that many of the objections are to cost benefit analysis generally and not about discounting as a distinct issue.⁸ Although we personally would use cost benefit analysis to evaluate projects, readers skeptical of this analysis should view our arguments as distinct from the debate on that issue. That is, we think that discounting is a related but distinct consideration from cost benefit analysis.⁹

How are we to evaluate this project? Our arguments below rely on the Pareto criteria – we argue that discounted cost-benefit analysis produces results that are Pareto superior to any other decision criteria, at least in most contexts. Pareto superiority is traditionally defined in terms of individuals; Pareto superiority would require that each and every individual throughout time is equal or better off with discounting. We apply the term to generations rather than individuals. We show below that each generation, however defined, is equal or better off. Particular individuals within a generation, however, may be better or worse off.

We take this approach for several reasons. First, the relevant actors in discounting debates are usually, if not always, generations. The debates are about justice between generations, not about redistribution within a

⁷There is an extensive literature on this topic. For an overview, see, The Journal of Legal Studies Volume 29 (2000).

⁸See e.g., Heinzerling [1998]; Ackerman and Heinzerling [2002]; McGarity and Shapiro [1996].

⁹In particular, it is possible to accept cost benefit analysis while rejecting discounting. Although few authors are explicit about this, it seems to reflects the views of lawyers such as Revesz [1999] and of the many economists who reject pure discounting such as Lind [1999], Cowen [1992], and Heal [1998]. Many discussions rejecting discounting, particularly in the legal literature, are found in broader analyses of cost benefit analysis and it is not clear whether the authors view the subjects as distinct. See, e.g., Ackerman and Heinzerling [2002; 1570-73]; McGarity and Shapiro [1996; 630].

generation. Moreover, each generation is, to a large extent, in charge of redistribution within that generation, perhaps favoring extensive redistribution, perhaps not. Given each generation more total resources makes whatever distributive decisions that want to make easier. Some might argue that we should not engage in a project that, relative to the status quo, changes the distribution of wealth with a generation, but this privileges the status quo. Moreover, it is not even clear in the abstract whether the alteration would be for the better or the worse or how it would interact with that generation's plans and ideals. Thus, we think the best approach is to think of Pareto superiority as applied to generations.

We can also frame the problem in the context of broader ethical principles. Schelling [1995] noted that the costs and benefits of long term projects are imposed on different individuals whose lives may be largely unconnected. For example, global warming abatement is likely to impose costs on the United States and create benefits for future generations of Indians, Chinese, and Bangladeshis. Similarly, using up resources today may transfer resources in the opposite direction, taking resources from those future foreigners and transferring it to today's Americans.¹⁰ The same is true if one analyzes projects with purely domestic consequences. One must still weigh the relevant costs and benefits to different, usually unrelated, individuals. Any analysis of the problem must take into consideration whether such a transfer is desirable. As Schelling [1995] argued, global warming abatement is as much a foreign aid problem as an environmental problem.

A welfarist consequentialist determines whether such a transfer is desirable by reference to the social welfare function. The transfer is desirable if social welfare increases because of the transfer. The difficulty with using such an analysis here is that we must decide who is in the social

¹⁰The whole notion of taking from one and giving to another requires a set of baseline entitlements which, in turn, may assume an ethical theory. We use this language and its implicit acceptance of the status quo for convenience and do not base our analysis on any assumption of baseline entitlements.

welfare function and at what weight. Utilitarianism and other forms of consequentialist reasoning do not answer this question directly. A broader ethical framework is needed.

The issues of transferring resources among nations and among generations are distinct. We tend to be much more altruistic to our heirs than to strangers. For example, a recent study shows that current U.S. foreign policies have an implicit weight on the life of a foreigner as low as 1/2000 of the life of an American. Kopczuk, Slemrod and Yitzhaki [2005] At the same time, most generations have left the world richer than they found it, implying that we value our heirs' lives highly. These behaviors, however, even if robust, may not comply with ethical precepts.¹¹

In this paper, we will weigh future lives the same as current lives. We assume the equal moral worth of all individuals, today and in the future. In particular, we assume that future lives enter into the social welfare function in exactly the same way current lives do. This means that we weight increases (and decreases) in well being of a future individual exactly the same as an equivalent increase (or decrease) in the well being of a currently living individual. We do not discount well-being.¹² We choose to weigh future lives equally because many of the discussions of discounting implicitly use this premise. Thus, as Fuchs and Zeckhauser [1987; 265] note, "[m]ost policy planning discussions assume full altruism

¹¹The same is true with respect to studies showing hyperbolic or other forms of irrational discounting of the future. Cropper and Laibson [1999]. Such studies do not show anything about ethical behavior toward the future.

¹²Many of the arguments against discounting use the rhetoric of "discounting lives." Heinzerling, [1999]. It seems, on its face, unethical to discount the life of some individual. Although this is a nice rhetorical trick, akin to relabeling the estate tax the death tax, it is not clear what it means. Advocates using this rhetoric would likely accuse us of discounting lives but our social welfare function weights future lives equally to current lives and future increases in welfare count exactly the same way current increases in welfare count. No life is being discounted.

- future citizens are given equal weight with present citizens." Pure equality may merely be a focal point, and it is not consistent with behavior with respect to cross-nation programs, but it is, nevertheless, likely to be the least objectionable choice.

Note that weighing individuals in a social welfare function equally does not mean equality of treatment. For example, if marginal utility declines with wealth or the social welfare function is concave, we will want to transfer resources from the rich to the poor.¹³ The reason is not unequal weighting. Instead, it is that overall social welfare may increase with such a transfer.¹⁴

This may be true with respect to projects. If the project transfers from the poorer to the richer, the decline in utility of the poorer may be more than the gain in the utility of the richer. Thus, even if the project passes discounted cost benefit analysis in absolute dollar terms, it may not increase social welfare. Similarly, if the project fails cost benefit analysis but redistributes from richer to poorer, it might increase social welfare. Every progressive tax system looks like this. Because of the deadweight loss of taxation, less money is transferred to the poor than is taken from the rich, but the tax may still be desirable.

A central point in our approach to the problem is the difference between discounting lives and discounting costs and benefits. As noted,

¹³The situation is even more complex than the text implies. It is possible to show that horizontal equity – equal treatment of equals – is *inconsistent* with welfarist consequentialism. [Atkinson and Stiglitz 1980; 354]

¹⁴While we do not address the appropriate degree of convexity of the social welfare function, note that there are arguments that convexity should be increase (possibly to infinity) if outcomes are uncertain (as opposed to merely risky in the Knightian sense). [Arrow 196_, others] Although we are not sure we believe that there Knightian uncertainty is a meaningful category, to the extent it is, very long-term projects of the type considered here are the most likely candidate. Because our arguments below are based on the Pareto criteria, we need not address this issue and, therefore, ignore problems created by Knightian uncertainty.

increments to well being are counted the same for current and future generations. Nevertheless, this does not mean that costs and benefits should not be discounted. Consider the principle example used by Revesz. Revesz [1999; 998]. He posits a two generation world where one generation lives from years 1 through 50 and the other from years 51 through 100. There are 100 units to divide between the generations. He assumes no productivity, which at least potentially translates into a zero discount rate, but suppose either that there is productive activity or there is otherwise an opportunity cost of funds, and that the discount rate is 3%.

Revesz posits that equal treatment of the two generations means that each should get 50 units of resources. He argues:

In the absence of discounting for time preference, each individual would be allocated 50 units of resources. In the face of a positive rate of time preference, however, even a relatively modest one, the first individual would get the bulk of the resources. It would be difficult to construct an attractive ethical theory that privileged the first individual in this manner merely because she lived fifty years earlier than the second individual.¹⁵

An alternative version of equality would give the first generation approximately 80 units and the rest to the second generation (giving each one an endowment of that amount in the first year of its life). When the second generation's endowment of 20 is invested at 3% for the first 50

¹⁵We should note that there is an important ambiguity in his argument. The example assumes zero productivity and, therefore, no opportunity costs of capital. He is explicitly discussing only pure time preference, which involves items other than pure opportunity costs, such as impatience, that might affect the discount rate. In such a case, our analysis is consistent with his. Revesz, however, believes that taking opportunity costs should be taken into account is different from discounting and that opportunity costs should be taken into account. Revesz [1999; 1007-1009]. If so, he should agree with the conclusions in the text, although it is not clear that he would. The example, therefore, by assuming a zero rate of productivity, while technically correct, had the potential to confuse.

years, it will grow to about 80 units in year 51, generating an equal division of the resources.

A social welfare function that weighs each generation equally would require the 80/20 split rather than the 50/50 split. The reason why is that with a 50/50 split, the second generation would have more total consumption. If its 50 unit endowment is invested until the time that generation is born, it will be able to consume about 220 units, compared to the 50 units consumed by the first generation. If equality is valued, there are welfare gains from transferring resources from the second generation to the first until the marginal utility of consumption of each generation is the same, which means that the endowment would be split 80/20. We adopt this latter concept of equality – intergenerational equity requires each generation to have the same marginal utility of consumption (or absolute utility, depending on our social welfare function). That is, we care about equality of well-being while those arguing against discounting are arguing for an equal division of current resources which will inevitably lead to an inequality in well-being.¹⁶

Much of the literature confounds these two topics, discounting individuals' lives and discounting costs and benefits, if not technically, at least rhetorically. For example, Cowen [1992] argues, using an axiomatic approach, that consequentialism implies a zero rate of intergenerational discount. Nothing in his argument, however, is inconsistent with discounting costs and benefits by their opportunity costs, and his claim that there should be a zero rate of intergenerational discount should not be taken to mean that costs and benefits of projects should not be discounted! All he is arguing for is that each generation have equal weight in the social welfare function, a result entirely consistent with the approach taken here. Similarly, concerns that discounting for time is equivalent to discounting the future lives are misplaced. The two concepts are distinct.

¹⁶There are a variety of ways we can view equality and the question "equality of what?" is important. See Sen [1992; 12-30]. We do not think, however, that Revesz's version of equality, equality of initial allocations, is a plausible version of equality.

II. Discounted cost benefit analysis Pareto dominants any other decision criteria

In this section, we show that discounted cost benefit analysis (unweighted for distributional issues) Pareto dominates any other decision criteria, at least under likely scenarios. Although these arguments are, to some extent, known, we lay them out in more detail that usual in order to explore the underlying assumptions. As noted, the arguments made here are purely arguments about opportunity costs. In the next section we discuss whether and the extent to which these argument are distinct from other types of discounting. We consider here two cases: first, a project that fails discounted cost benefit analysis but that seems attractive for distributional reasons and second, a project that passes cost benefit analysis but that seems unattractive for distributional reasons.

Consider the case where a project fails discounted cost benefit analysis. That is, the discounted marginal costs exceed the discounted marginal benefits. The project may still improve social welfare because of the distribution of the costs and benefits. For example, it might impose costs on the rich and endow the poor with benefits. Nevertheless, we should not engage in the project because there will always be another project that is a Pareto improvement.

Suppose, for example, that a project will impose costs today and help a future generation. Given a valuation of life for today and in the future (which may differ because willingness to pay for life may increase as wealth increases), suppose that the discounted future increase in welfare is less than the current decrease. For example, suppose we were to spend \$1 million to save \$100 million in one hundred years. If the discount rate is more than (roughly) five percent, this project fails discounted cost benefit analysis because the present value of the \$100 million benefit is less than the \$1 million cost.

If we were considering this project, we would be better off engaging in other projects that meet discounted cost-benefit analysis. For example,

Paretian Intergenerational Discountin	g Page 13

we could take the present value of \$100 million and invest it in a project that receives the market rate of return. Future generations would still get \$100 million but the current cost to us would be lower. Alternatively, we could invest the full \$1 million, imposing the same cost on the current generation, and give future generations more. Either way, at least one generation is better off and nobody is worse off. Therefore, regardless of whether the project increases welfare, we should not engage in the project because there are other projects that increase welfare more. If it were the case that other projects were limited, the case might be different but we can always invest funds at the market rate.

This argument gives an additional reason why the claim that discounting does not respect future generations as equals is incorrect. Suppose that we decide the engage in this project on the basis that future generations are equals and, because \$100 million is more than \$1 million, the project is worthwhile. If we could ask future generations whether they would want us to engage in this project, they would prefer that we just invest the money at the market rate of return because they would be better off with such an investment. If respecting future generations means anything, it should mean respecting our best guess as to their wishes and helping them as much as feasible. A project that fails discounted costbenefit analysis fails this minimal test. Rabl [1996] makes a similar argument about the desires of future generations.

The other case is projects that pass discounted cost benefit analysis but the proposed decision criteria would recommend rejecting them. This might be the case because the project has bad distributional consequences. For example, it might create benefits for the rich that exceed on a dollar basis the costs it imposes on the poor but, because the poor feel the pain of the loss of a dollar more than the rich feel the benefit of the gain, the project, on its own reduces social welfare. Rejecting this project, however, would be a Pareto dominated decision.

Suppose that the project produces benefits of \$100 and only costs \$90, both in present value terms. In addition, suppose that the benefits go

Samida and Weisbach	Page 14
	145011

to wealthy individuals who, because of declining marginal utility of wealth, do not value them very much, and the costs are imposed on poor individuals who view the costs as a large imposition. Although the project taken alone would seem to reduce welfare, there exists an offsetting transfer that makes the project a Pareto improvement. In particular, the wealthy individuals would be willing to give up \$95 to get the project (gaining \$5) and the poor would be willing to accept \$95 to agree to the project (gaining \$5). Thus, although the project taken alone reduces welfare, the combination of the project plus the transfer is Pareto improvement over rejecting the project.

Future generations, of course, are not yet alive to strike such a bargain with us and even if they were alive, no such bargain would be enforceable. And it is difficult to even imagine a bargain because the ethical discussion is about the rights and responsibilities of each generation and without a background set of rights and responsibilities, there cannot easily be a bargain.¹⁷ Without an actual transfer, the project is merely Kaldor-Hicks efficient and may not be consistent with improving social welfare.

By hypothesis, there is a project on the table that has costs and benefits to the present and the future. The current generation is entirely in control of the project – it can decide whether to engage in it entirely without regard to the future.¹⁸ Given whatever level of altruism (or selfishness) the current generation has, however, it makes sense to exercise that altruism in the most efficient fashion. That is, it never makes sense to throw resources away. Therefore, there need not be a bargain

¹⁷See Sunstein and Rowell [2005; 16]. This is not necessarily the case. Individuals frequently bargain successfully in the presence of great uncertainty about their underlying rights and responsibilities.

¹⁸Although moral theories can reject, as an ethical matter, the dictatorship of the present, they cannot reject it as a factual matter. The current generation can make the decision whether to engage in the project or not.

between generations or an enforcement mechanism for the Pareto dominance argument to hold.

To illustrate, go back to our example, and suppose that the current generation is rich and the future expected to be poor, and that, relative to the status quo, the project transfers \$100 to the current generation at the expense of \$90 to the future. Suppose also that the current generation otherwise plans to transfer \$1,000 to the future because of its altruism. The current generation should engage in the project and increase its transfer to the future by somewhere between \$90 and \$100. The current generation is better off – it meets its altruistic or selfish goals (depending on what one thinks about the transfer of \$1,000) and gets the project. The future generation is also better off. There is no need for an actual bargain for the transfer to occur. The current generation could always do the project without the transfer of \$90 to the future, but it could also just reduce the legacy it would otherwise leave from \$1,000 to \$990.

We can produce similar examples where the current generation is poor and the future rich or where the costs and benefits of the project go in different directions or where the level of altruism changes. As long as there are sufficient transfers among generations, these existing transfers can be altered to make the project a Pareto improvement if it passes discounted cost benefit analysis.

The argument is closely related to a theorem from the public finance literature first put forth by Atkinson and Stiglitz [1976]. They considered the case of optimal redistributive tax and transfer systems but their argument has been extended to cost benefit analysis and to public goods.¹⁹

¹⁹Legal readers are likely to know this theorem from the arguments about the desired efficiency of legal rules. Shavell [1981]; Kaplow and Shavell [1994]. The tax version of this argument has been extensively discussed in the literature. See Deaton [1981]; Stiglitz [1987]; Saez [2002]; Kaplow [2004]. The application to cost benefit analysis was first suggested by Hylland and Zeckhauser [1979]. Kaplow [1996] considers the application to public goods.

These extensions show that cost benefit analysis should not, under the conditions of the theorem, be adjusted for distributional considerations, regardless of how egalitarian we are. Hylland and Zeckhauser [1979]. Similarly, the government should purchase public goods without regard to the tax cost of financing them. Kaplow [1996].

We can use the same example used above to illustrate the argument.²⁰ In the example, rich individuals were made better off by \$100 and the poor worse off by \$90. Rather than a voluntary transfer from the rich to the poor to pay for the project, we would use an identical adjustment to the tax schedule. Suppose we increase the progressivity of the tax system by increasing taxes on the rich by \$100 and reducing taxes on the poor by \$90, leaving both classes of individuals in the same place as without the project and tax adjustment. There are no distributive effects of the combined project and tax adjustment. Unlike with the voluntary exchange case considered above, however, taxes might create deadweight losses, such as those caused by a reduction in labor effort. Note, however, that by leaving each individual in the same place, the combination of the project and tax adjustment introduces no distortions into the economy, such as the usual deadweight loss caused by a tax. Finally, because taxes went up on the rich by \$100 and down on the poor by only \$90, there is an additional \$10 of tax receipts that can be used to make everyone (or some people) better off, resulting in a Pareto improvement.²¹ The

²⁰The version of the argument here is due to an interpretation of the Atkinson Stiglitz 1976 result originally by Hylland and Zeckhauser [1979] and treated extensively in Kaplow and Shavell [1994]; Kaplow [1996], and Kaplow [2004].

²¹Sophisticated readers are probably thinking at this point of the theorem of the second best, which states that distortion counting of the sort in the text cannot be used for welfare analysis. One of their assumptions, namely weak separability between labor and leisure eliminates this problem. Relaxing this assumption means that it is possible that the tax on the commodity reduces the distortion in labor supply caused by the labor income tax. The commodity, for, example, might be a relative complement to leisure. In practice, the possibility of taxing relative complements to leisure is obscure and has few if any implications for intergenerational projects. For a discussion, see Weisbach and Bankman, [2005].

conclusion is that as long as the tax system is sufficiently flexible to make offsetting distributional adjustments, projects should be undertaken if and only if they pass unweighted cost benefit analysis.

This conclusion, usually stated for projects with present costs and benefits, applies to projects with costs and benefits spread over time and between generations. As long as the discounted costs are less than the discounted benefits, there exists an offsetting tax and transfer adjustment that makes everyone better off, even if the project taken alone seems to have bad distributional effects. Therefore, the appropriate decision criteria for projects is discounted cost benefit analysis.²²

II. Counter-arguments

The Pareto dominance arguments have long been recognized in the literature, but, we believe, have not been uniformly accepted because of a variety of objections. We consider some of these objections here.

A. Intervening Generations

One of the most common counter-arguments to the Pareto dominance argument is that intervening generations may abscond with any

²²There is a subtlety. In Hylland and Zeckhauser [1979] or particularly Kaplow [2004], the tax adjustment offsets the public good, excise tax, project or the like. The tax will have an affect on the margin but this is offset by the benefit (or cost) of the project on the margin. For example, suppose that the benefit of the project goes up linearly with income. Increasing the income tax to the same extent will not distort labor effort because if an individual works harder he gets both the benefit of increased enjoyment from the project and increased taxes. In the intergenerational context, both the tax and the project costs and benefits are more lump sum. For example, suppose that the current generation must transfer \$100 to a future generation because a project benefits us by \$100. Assuming that the transfer will be made, the tax cannot be avoided by working less or the like and the same goes for the benefit of the project.

savings that we intended to leave for distant future generations.²³ Lind [1999; 174-175] is a particularly emphatically stated version. He argues:

[I]ntergenerational transfers, either forward or backward, generally must be made through series of intervening generations. Designing and implementing such transfers is virtually impossible, and the potential for an intervening generation to break the chain of transfer makes such transfer schemes virtually impossible to implement.

Lind goes on to dramatize the point by hypothesizing a proposal to transfer resources to the future through an investment with a zero percent rate of return (at a time when money or other projects earn a 10% return). Lind [1999; 176] argues

The preferred decision may well be to make that investment and transfer the resources to the future generation even though it earns a zero rate of return. At this point an eager graduate student jumps up, sensing an economic slam dunk, and says, "that was a really dumb decision. You could have invested that money at 10% and made those people a lot better off." Wrong! We don't know how to set aside investment funds and to commit intervening generations to investment and reinvesting those funds for eventual delivery as consumer goods to the generation 200 years from now.

It is Lind, however, and not the eager graduate student who is wrong. The reason why is that any project, including the project being analyzed such as Lind's zero rate of return project, can be destroyed by an intervening generation. Absent a reason to believe that transfers from one project are less easily misdirected than transfers from others, we should pick the project with the highest rate of return.

²³See, e.g., Sunstein and Rowell; draft at 14 for a version of this argument.

Paretian Intergenerational Discounting	Page 19

For example, suppose we are considering conserving petroleum for use in 200 years, and the rate of return on this project is less than the market rate of return. We should not engage in the project because other projects would confer more value on future generations for less money. If we are worried about intervening generations with respect to the alternative projects, we have to worry that intervening generations will interfere with the project at issue, say by using up the petroleum themselves. Similarly, if we are considering transferring resources to a future generation by abating global warming, an intervening generation could always take the transfer for itself by releasing excess carbon into the air, destroying our attempt to preserve the resource for the future.

Lind and others who make this claim, however, do not make the necessary institutional arguments about the ability of future to interfere with some transfers but not others. They give no more reason to believe that intervening generations will respect an environmental choice than they would leave money in the bank or in another type of project. They assume, nonetheless, that this would be the case – environmental projects (or whatever unstated projects are uppermost in the analyst's mind) are assumed to be feasible methods of transferring resources while other projects with a market rate of return are assumed to be infeasible methods of transferring resources. While it is possible (indeed likely) that some transfers are more easily reversed than others, such a claim is not sufficiently general to reject discounting. Arguments about which transfers cannot be undone are likely to be very project specific and do not present any general theory.²⁴

Moreover, it is not clear that tying the hands of intervening generations would be wise. We can imagine that intervening generations are more, equal to, or less altruistic to the future than we are. If they more than or equal to the current generation, Lind's argument presents no issue. If they are less altruistic, we would have to ask why. It could be the case

²⁴Arrow [1999; 18] makes a similar point.

Samida and Weisbach	Page 20
	1 uge 20

that they have discovered new and better ethical theories and are behaving consistently with those theories. We should probably want to respect that choice. Alternatively, they might merely be selfish or have adopted worse ethical theories in which case we might want to tie their hands. If we can successfully tie the hands of future generations (by engaging in one particular project but not in others), however, we would prevent both the unwise generation from reneging and the wise generation from improving. Thus, even if we could establish, on a project by project basis which ones are less subject to reneging, it is not clear that engaging in those projects would be wise.

The discounted cost benefit analysis assumes that transfers to or from the future can be made at the market rate of return. This is a good first-best baseline from which to think about the issue. To incorporate Lind's arguments, future work would have to make institutional or second-best arguments, trying to distinguish which types of projects are better or worse from a commitment standpoint. We see no particular reason why such an analysis would favor the types of long-term projects currently discussed, such as global warming abatement or prevention of species extinction, although it might. (To be sure, these projects may very well pass discounted cost benefit analysis in any event. We take no position on that issue.)

B. Difficulty of transfers

A common reaction to the Atkinson/Stiglitz tax argument when applied to projects is that the scope of most projects is too narrow (and the scope of taxation too broad) for such adjustments to be made. For example, we might be deciding whether to build a highway that produces noise and pollution next to a poor neighborhood and benefitting those rich who happen to use it as their commuting route. Broad based tax adjustments will not be able to tax those particular rich and benefit those particular poor affected by the project. But if we are talking about large scale intergenerational issues – the only kind worth our attention – this objection goes away. The relative redistribution is between current and future generations rather than particular individuals. (To be sure, even large scale projects will have narrow effects. The costs of reducing global warming will not fall on all Americans equally. But intergenerational equity is not usually concerned with these intragenerational issues.)

Another objection frequently made is that the offsetting tax adjustments might not be made, for purely political reasons such as the separation of powers in the U.S. system. For example, Adler and Posner [2000] argue that an agency deciding whether to pursue a project does not have the choice of making offsetting tax adjustments. It is hard to see why the adjustments would not be made, however, at least on an economywide, systematic basis. Positing that projects will have good distributional effects but taxes cannot be adjusted to have the same effects requires very strong assumptions about the political process, assumptions that are unlikely to hold systematically. Instead, it is better to assume that the political process produces some equilibrium amount of redistribution, which may or may not be optimal. More importantly, these sort of objections have no bearing on large-scale intergenerational projects. Any project that has sufficient redistributive impact to be worth our attention is going to have legislative involvement, and the legislature clearly has power over both the project and taxes.

The main assumption made by the theorem that might affect its application to intergenerational projects is that it assumes that the tax and transfer system can be adjusted to offset the effect of the project. If a project is going to impose costs today and produce benefits in 200 years, we must be able to adjust taxes and transfers between these two generations to offset any distributional implications. Leaving aside the "intervening generation" problem, discussed above, there should clearly be sufficient intergenerational transfers that can be adjusted. Suppose, for example, we are going to incur costs today to help those in 200 years and we believe that individuals in 200 years will be systematically better off. The project would redistribute in the wrong direction, helping those who are better off. To offset this redistribution, the project could be deficit financed or other taxes could be reduced. The legacy each generation leaves to the next is enormous, and there is no reason to believe that adjustments to offset bad distributional consequences of projects would not be available.

The most serious objection to the above analysis is that the tax and transfer system is not sufficiently flexible to transfer resources to or take resources from future generations of individuals in other nations. For example, abating global warming may impose costs on the current U.S. generation and help future Chinese. This seems a valid objection. There are ways to redistribute to and from future foreigners by adjusting foreign aid, military spending, and the like, but there nothing very direct and efficient.

If we were to weight future foreigners in our social welfare function at the same weight as current locals and if a given project is the best way of transferring resources to poor future foreigners, it would be welfare improving to adopt the project. For example, the argument would be that the United States should incur costs today to abate global warming because doing so is the best way to help future foreigners. That is, those arguing against discounting must directly confront Schelling's [1995] that most long-term projects are really foreign aid projects.

As noted, foreigners are barely weighted in a derived U.S. social welfare function. Kopcuk, Slemrod, and Yitzhaki [2005]. If we weight foreigners equally, current projects would have to be changed so dramatically that most long-term projects, such as global warming abatement, would no longer be near the top of our priority list. Alternatively, if we take our current valuation of foreigners, we would vastly discount their lives in our social welfare function and most projects of a global scale such as global warming abatement would not remotely pass cost-benefit analysis.

For example, suppose we are considering a project that will cost the current generation living in the United State \$100 and help a future generation of poor foreigners by a present value amount of \$90, and that

Paretian Intergenerational Discounting	
--	--

because of distributional considerations, the benefits to the future foreigners are more than costs to current U.S. lives. Our argument above was that the project should be rejected because it is better to transfer \$90 directly. Suppose, however, that we would never transfer the \$90, for either good or bad reasons. It makes no sense, in such a case, to spend \$100 to transfer the \$90. To suggest that we really do want to transfer the \$90 and that the only way to do so is to engage in a project costing \$100, while possible, seems implausible.

Page 23

Another way to look at this issue is to compare the Pareto analysis with a welfarist analysis. The example immediately above was a Pareto analysis. For any given level of transfers to foreigners, it should be done efficiently. A welfarist would not be happy with this because a welfarist would insist on the right level of transfers to foreigners, a level likely to be significantly higher than current transfers. Such a welfarist would might then have a valid argument that the distributive effects of the project must be considered rather than merely using discounted cost-benefit analysis. At the same time, however, the welfarist might be more concerned about our current level of foreign aid or the current distribution of resources in the world.

C. Difference between opportunity costs and discounting

Many commentators have argued that taking opportunity costs into account is not the same as discounting. Others have argued that it is not appropriate to take opportunity costs into account. We consider these arguments in this section.

Taking opportunity costs into account presumably means taking into account the alternative uses of resources that we are considering using for a project. This is equivalent to taking the future value of these resources at the highest rate of return available. There is, however, a perfect mathematical equivalence between discounting and taking future values. Algebraically, the two are the same, and any project that passes (fails)

discounted cost benefit analysis passes (fails) future valued cost benefit analysis.

Everyone knows this, so commentators arguing that opportunity costs should be taken into account but projects not discounted must have something else in mind. A significant possibility is that they are focusing on reasons for discounting other than the opportunity cost of capital. For example, some have suggested that interest rates are positive because of impatience (pure time preference) or because of the possibility of a catastrophe (better to consume the resources than have them destroyed in the catastrophe). In this paper we do not consider these rationales for discounting and, therefore, need not consider whether they are appropriate. Note that this means the discounting argued for here would be at a lower rate than the market rate of interest because the market rate of interest will reflect factors other than opportunity costs. Rabl [1996] for example, argues one these grounds for a long-term discount rate that reflects the expected growth in the economy than recently observed interest rates. Weitzman [2001] makes a similar argument.

Parfit [1984; 483] gives four reasons why the existence of opportunity costs does not support discounting. First, he argues that some benefits are consumed rather than reinvested. He posits the enjoyment each year of a stretch of countryside, arguing that enjoyment in later years should not count less than enjoyment in earlier years. Consumption, however, can be thought of as a stream of money – the willingness to pay or accept. If the value to an individual of the countryside in 10 years of \$100, the individual would accept less than \$100 today to give up the enjoyment of the countryside in 10 years. There is no difference between consumption items that are consumed and those that are reinvested.

Broome [1994; 148] draws a conclusion similar to Parfit's from the same example, although for different reasons. Broome generally favors discounting on a theory closely analogous to the opportunity-cost arguments made here. Broome argues that because of the productivity of capital, most commodities will be cheaper in the future – this is why there

is a positive rate of interest and why commodities should be discounted. For example, producing televisions gets cheaper, so the market prices today of televisions today and in the future are different. He uses the idea of a commodity's "own interest rate" as the rate of change of the price of the commodity. One should agree to pay less for future televisions than today's televisions because of this "fertility" of capital. Open countryside, however, is not fertile. "Scarce resources cannot be converted into a greater quantity of future resources, and they therefore have own interest rates of zero or thereabouts." Broome [1994; 148] If the value of a commodity will remain the same through time, it should not be discounted.

This argument is consistent with the opportunity costs arguments made here. If the absolute value of the scarce resource remains the same but other resources get cheaper, the relative price of the scarce resource will go up. Indeed, this is why we can think of there being a single interest rate in the economy notwithstanding each commodity having its own "fertility." Therefore, one can reframe Broome's argument as an argument that a consistent opportunity-cost discount factor should be used but that it should be based on relative future prices. If the price of the scarce resource goes up at the same rate as the discount factor, the net effect of determining the future price and discounting it back is zero. If the price of the resource goes up faster than the discount rate, the net effect is like having a negative discount factor.

Does it matter which procedure is used – not discounting (or more precisely determine a commodity's own rate of interest) or determining future prices and discounting, if the net effect is the same? In a technical sense no – all we care about is making the welfare maximizing decisions. Rhetorically, however, commentators such as Parfit use these sorts of

arguments as a wedge to argue against discounting more generally and it is important to realize that they are not inconsistent with discounting.²⁵

Parfit's second argument is that many harms will not be compensated. He considers the case of a pollutant that will cause congenital deformities among some unlucky future individuals and assumes that they will not be compensated. There are two distinct elements in the example. The first is similar to the countryside example above. A deformity tomorrow reduces welfare just as much as deformity in the future, so deformities may not be like commodities that get cheaper over time. The same answer given above applies and, in fact, the case might be easier, because the cost of care of or cure for deformities is likely to go down over time, implying that there is an "own interest rate." Broome [1994; 149-150] Second, there is the issue of non-compensation. This argument does not distinguish opportunity costs from discounting more generally. The arguments about the need for compensation and the likelihood of providing compensation were discussed above. It is our view that for large scale projects, compensation can be provided, at least for within-country effects.

Third, Parfit argues that in some time periods, there will be no opportunity costs because investment will bring no return. While this is

²⁵Moreover, Broome's underlying assumption, that some goods are scarce and, therefore, their price will go up over time at least as fast as the overall discount rate, is also controversial. Julian Simon and his critics have been debating this for more than 30 years without apparent resolution. Even if one disagrees with Simon, and many do, at a minimum Simon is correct that it is difficult to make a prediction like Broome's, that the price of a given scarce commodity will go up over time. For example, Parfit and Broome use open countryside as their example. It is difficult to determine the price of open countryside because the term has no precise meaning. On possibility is that open countryside is any land that is neither urban nor farms. Data show that the amount of non-urban, non-farm land in the continental U.S. has increased in the last half century. Since 1945, urban areas have grown by roughly 69 thousand square miles. Since 1950, farmland has decreased by almost 360,000 square miles. See http://www.demographia.com/db-1945uza.htm and

http://www.usda.gov/nass/pubs/trends/farmnumberslandsize.csv

possible, it is a technical economics question to estimate the likely opportunity costs, not an ethical question about whether discounting is appropriate in the presence of those costs.²⁶

Finally, Parfit argues that the rhetoric of discounting is inappropriate and that a project's costs and benefits should be expressed in a "more temporally neutral way." Parfit [1984; 484]. This same idea is echoed in Heinzerling [1999;] and Revesz [1999; 1008] For example, Heinzerling prefers the language of opportunity costs and future values because it "elides normative issues." Revesz claims that discounting the utilities of future generations is the objective function of a specific ethical theory while "paying attention to opportunity costs does not imply the choice of any particular theory." We fail to see how one mathematical procedure can present different moral issues than another identical mathematical procedure. If the issue is about presenting information to decision makers or the public in a way that helps inform them, we are in favor of such a presentation, but nothing in these discussions makes a serious case that one form of presentation is better in this regard than another. Our intuition is that discounting is likely to be more informative because it presents all information using current dollars which policymakers can easily understand.

In sum, the only merit to the position that discounting and taking opportunity costs into account are not the same is with respect to reasons for discounting unrelated to the productivity of capital, such as impatience. Claims that taking opportunity costs into account is different from discounting are erroneous.

²⁶Cowen and Parfit (1992) argue that the discount rate is a choice variable in the decision because it depends on the extent of savings. Suppose that we decide to pass more on to future generations by saving more. If the marginal product of capital is declining, this will reduce the discount rate. They conclude that this means that we should not discount future prices to the present when making decisions. We agree with Broome's [1994; 140] explanation of why this is not the case.

III. Additional Issues

There is a large technical literature on determining the correct discount rate. Issues arise such wedges between the return to consumption and the return to investment, adjusting for risk, and estimating future productivity. For example, Weitzman [2001] presents arguments for a low discount rate based on how one aggregates the probabilities of various discount rates in the future. For a summary, see Tresch [2002; 733-756] These arguments are very important – changing the discount rate even a small amount can change whether many projects pass cost benefit analysis. Nevertheless, they are technical in nature – they do not have to do with the propriety of discounting.

Another issue is whether it is appropriate to discount life. Many commentators object to discounting life but not money. Revesz [[1999,__]; Heinzerling [date]. As should be apparent from the discussion so far, we do not believe that discounting for opportunity costs involves discounting future lives. Future lives are weighed equally to current lives and a loss of a future life reduces social welfare just like loss of a current life. What we propose to discount is expenditures and receipts, or in the case of projects that directly affect utility without market intermediation, the monetary equivalents. As Sunstein and Rowell [2005] emphasize, harms to lives, whether present or future, can be and should be thought of in monetary terms.

Another common argument is that discounting is inappropriate when there are significant irreversibilities, such as in the case of global warming or extinction of a species. The right approach to irreversibilities, however, is not to refuse to discount. Instead, it is to use real option theory. This theory takes into account the problem with lost opportunities or sunk investments. It also fully incorporate discounting (necessarily so because it is attempting to model when to incur an irreversible cost). Although we must be cautious about irreversible actions, refusing to discount is unlikely to get the right result.

1 di chun inici gener anonai Discounting 1 dge 2	Paretian Intergenerational Discountin	g Page 29
--	---------------------------------------	-----------

Finally, we should say a word about rhetoric. Because of the power of discounting it is easy to come up with rhetorical quips that make discounting seem unattractive. Ackerman and Heinzerling object that with "a discount rate of five percent, for example, the death of a billion people 500 years from now becomes less serious than the death of one person today." Ackerman and Heinzerling [2002; 1571]. As noted, our favorite is Cowen and Parfit's [1992; 145] statement, "Imagine finding out that you, having just reached your twenty-first birthday, must soon die of cancer because one evening Cleopatra wanted an extra helping of dessert."

Many of the claims comparing stubbing one's toe to blowing up the world implicitly depend on discount rates over long periods of time that are far higher than possible. If the discount rate used is greater than the long run growth rate of the economy, a project of any size will, with sufficient time, be worth more than all the wealth in the world, an obvious contradiction (the project would be part of the world so it could not be worth more than the world). Under the argument we make here, long-run discount rates should instead be limited to the expected long-run economic growth because this is the true opportunity cost. For a similar argument, see Rabl [1996]. With this more appropriate discount rates, current costs will not grow to implausible sizes in the long-run future. If a cost is, say, 1% of the current economy, discounting will keep it at 1% of the future economy. Stubbing one's toe remains stubbing one's toe.

Moreover, what these quips miss is that discounting provides for Pareto superior outcomes. Failure to make decisions following discounted cost benefit analysis means throwing resources away to nobody's benefit. For example, the reverse Cowen and Parfit [1992] problem arises if one looks forward rather than backwards. Suppose that you are going to spend a dollar today. That dollar will grow to a billion, a trillion or more dollars with enough time. Spending that dollar today means taking that money from the future. To paraphrase Cowen and Parfit, every time you eat a banana, you condemn a million people in the future to death.

IV. Conclusion

We conclude that discounting by the opportunity cost of capital is generally appropriate and Pareto dominates any other decision procedure. The most difficult case is where a project transfers significant amounts to or from future foreigners and where it is difficult make those transfers or offsetting transfers in any other way.

References

Matthew D. Adler and Eric A. Posner, Implementing Cost-Benefit Analysis when preferences are distorted, 29 J. Legal Studies 1105-1147 (2000).

Frank Ackerman and Lisa Heinzerling, Pricing the Priceless: Cost-Benefit Analysis of Environmental Protection 150 U. Penn. L. Rev. 1553 (2002).

Kenneth Arrow, at all, 1996. Intertemporal Equity, Discounting and Economic Efficiency, in Climate Change 1995: Economic and Social dimensions of Climate Change 179 (James p. Bruce, at all eds., 1996).

Atkinson, Anthony B. and Joseph E. Stiglitz. 1976. The Design of Tax Structure: Direct versus Indirect Taxation, 6 Journal of Public Economics 55-75.

Anthony B. Atkinson and Joseph E. Stiglitz, Lectures on Public Economics (1980).

Kenneth J. Arrow, Discounting, Morality, and Gaming, in Discounting and Intergenerational Equity (Paul R. Portney and John P. Weyant, eds 1999).

Joseph Bankman and David A. Weisbach, The Superiority of an Ideal Consumption Tax over an Ideal Income Tax, working paper (2005).

David Bradford, On the Uses of Benefit-Cost Reasoning in Choosing Policy toward Global Climate Change, in Discounting and Intergenerational Equity (Paul R. Portney and John P. Weyant, eds. 1999).

John Broome, Discounting and Welfare, 23 Philosophy and Public Affairs 128-156 (1994).

Graciela Chichilnisky, An axiomatic approach to sustainable development, 13 Social Choice and Welfare 231 (1996).

Tyler Cowen, Consequentialism Implies a Zero Rate of Intergenerational Discount, in Justice between Age Groups and Generations (Peter Laslett and James S. Fishkin eds. 1992).

Tyler Cowen and Derek Parfit, Against the Social Discount Rate, in Justice between Age Groups and Generations (Peter Laslett and James S. Fishkin, eds 1992).

Maureen Cropper and David Laibson, The Implications of Hyperbolic Discounting for Project Evaluation, in Discounting and Intergenerational Equity (Paul R. Portney and John P. Weyant, eds. 1999).

Victor R. Fuchs and Richard Zeckhauser, Valuing Health – A "Priceless" Commodity, 77 American Economic Review 263-268 (1987).

Geoffrey Heal, Economic Theory and Sustainability (1998).

Lisa Heinzerling, Discounting Life, 108 Yale L. J. 1911 (1999).

Hylland, Aanund and Richard Zeckhauser. 1979. Distributional Objectives Should Affects Taxes but Not Program Choice or Design, 81 Scandinavian Journal of Economics 264-283.

Kaplow Louis. 2004. On the Undesirability of Commodity Taxation even when Income Taxation is not Optimal, Harvard John M. Olin Center for Law and Economics Discussion Paper No. 470.

Louis Kaplow, The Optimal supply of Public Goods and the distortionary costs of taxation, 49 National Tax Journal 513-533 (1996).

Kaplow, Louis, and Shavell, Steven. 1994. Why the Legal System is less Efficient than the Income Tax in Redistributing Income, 23 J. Legal. Studies 667-__.

Wojciech Kopczuk, Joel Slemrod, and Shlomo Yitzhaki, The limitations of decentralized world redistribution: An optimal taxation approach, 40 European Economic Review 1051-1079 (2005).

Robert C. Lind, Analysis for Intergenerational Decisionmaking, in Discounting and Intergenerational Equity (Paul R. Portney and John P. Weyant, eds. 1999).

Thomas O. McGarity and Sidney A. Shapiro, OSHA's Critics and Regulatory Reform, 31 Wake Forest L. Rev. 587 (1996).

Derek Parfit, Reasons and Persons (1984).

Paul R. Portney and John P. Weyant, Introduction, inDiscounting and Intergenerational Equity (Paul R. Portney and John P. Weyant, eds 1999).

Eric A. Posner and Cass R. Sunstein, Dollars and Death, 72 U. Chicago L. Rev. 537 (2005).

Ari Rabl, Discounting of long-term costs: What would future generations prefer us to do? 17 Ecological Economics 137-145 (1996).

Richard L. Revesz, Environmental Regulation, Cost-Benefit Analysis, and the Discounting of Human Lives, 99 Colum. L. Rev. 941-1017 (1999).

Thomas C. Schelling, Intergenerational discounting, 23 Energy Policy 395-401 (1995).

Amartya Sen, Inequality Reexamined (1992).

Stiglitz, Joseph E. 1982. Self-Selection and Pareto Efficient Taxation, 17 Journal of Public Economics 213-240.

Stiglitz, Joseph E. 1987. Pareto Efficient and Optimal Taxation and the New New Welfare Economics, in 2 Handbook of Public Economics 991-1042 (Alan J. Auerbach and Martin Feldstein eds.).

Cass Sunstein and Arden Rowell, On Discounting Regulatory Benefits; Risk, Money, and Intergenerational Equity, draft.

Richard W. Tresch, Public Finance, A Normative Theory (2002).

Martin Weitzman, Gamma Discounting, 91American Economic Review 260-271 (2001).

29 The Journal of Legal Studies (2000).

Readers with comments should address them to:

Professor David A. Weisbach University of Chicago Law School 1111 East 60th Street Chicago, IL 60637 d-weisbach@uchicago.edu

Chicago Working Papers in Law and Economics (Second Series)

- 1. William M. Landes, Copyright Protection of Letters, Diaries and Other Unpublished Works: An Economic Approach (July 1991)
- 2. Richard A. Epstein, The Path to *The T.J. Hooper*: The Theory and History of Custom in the Law of Tort (August 1991)
- 3. Cass R. Sunstein, On Property and Constitutionalism (September 1991)
- 4. Richard A. Posner, Blackmail, Privacy, and Freedom of Contract (February 1992)
- 5. Randal C. Picker, Security Interests, Misbehavior, and Common Pools (February 1992)
- 6. Tomas J. Philipson & Richard A. Posner, Optimal Regulation of AIDS (April 1992)
- 7. Douglas G. Baird, Revisiting Auctions in Chapter 11 (April 1992)
- 8. William M. Landes, Sequential versus Unitary Trials: An Economic Analysis (July 1992)
- 9. William M. Landes & Richard A. Posner, The Influence of Economics on Law: A Quantitative Study (August 1992)
- 10. Alan O. Sykes, The Welfare Economics of Immigration Law: A Theoretical Survey With An Analysis of U.S. Policy (September 1992)
- 11. Douglas G. Baird, 1992 Katz Lecture: Reconstructing Contracts (November 1992)
- 12. Gary S. Becker, The Economic Way of Looking at Life (January 1993)
- 13. J. Mark Ramseyer, Credibly Committing to Efficiency Wages: Cotton Spinning Cartels in Imperial Japan (March 1993)
- 14. Cass R. Sunstein, Endogenous Preferences, Environmental Law (April 1993)
- Richard A. Posner, What Do Judges and Justices Maximize? (The Same Thing Everyone Else Does) (April 1993)
- 16. Lucian Arye Bebchuk and Randal C. Picker, Bankruptcy Rules, Managerial Entrenchment, and Firm-Specific Human Capital (August 1993)
- 17. J. Mark Ramseyer, Explicit Reasons for Implicit Contracts: The Legal Logic to the Japanese Main Bank System (August 1993)
- William M. Landes and Richard A. Posner, The Economics of Anticipatory Adjudication (September 1993)
- 19. Kenneth W. Dam, The Economic Underpinnings of Patent Law (September 1993)
- 20. Alan O. Sykes, An Introduction to Regression Analysis (October 1993)
- 21. Richard A. Epstein, The Ubiquity of the Benefit Principle (March 1994)
- 22. Randal C. Picker, An Introduction to Game Theory and the Law (June 1994)
- 23. William M. Landes, Counterclaims: An Economic Analysis (June 1994)
- 24. J. Mark Ramseyer, The Market for Children: Evidence from Early Modern Japan (August 1994)
- 25. Robert H. Gertner and Geoffrey P. Miller, Settlement Escrows (August 1994)
- Kenneth W. Dam, Some Economic Considerations in the Intellectual Property Protection of Software (August 1994)
- 27. Cass R. Sunstein, Rules and Rulelessness, (October 1994)
- 28. David Friedman, More Justice for Less Money: A Step Beyond *Cimino* (December 1994)
- 29. Daniel Shaviro, Budget Deficits and the Intergenerational Distribution of Lifetime Consumption (January 1995)
- 30. Douglas G. Baird, The Law and Economics of Contract Damages (February 1995)
- Daniel Kessler, Thomas Meites, and Geoffrey P. Miller, Explaining Deviations from the Fifty Percent Rule: A Multimodal Approach to the Selection of Cases for Litigation (March 1995)
- Geoffrey P. Miller, Das Kapital: Solvency Regulation of the American Business Enterprise (April 1995)
- 33. Richard Craswell, Freedom of Contract (August 1995)
- 34. J. Mark Ramseyer, Public Choice (November 1995)
- 35. Kenneth W. Dam, Intellectual Property in an Age of Software and Biotechnology (November 1995)
- 36. Cass R. Sunstein, Social Norms and Social Roles (January 1996)

- J. Mark Ramseyer and Eric B. Rasmusen, Judicial Independence in Civil Law Regimes: Econometrics from Japan (January 1996)
- Richard A. Epstein, Transaction Costs and Property Rights: Or Do Good Fences Make Good Neighbors? (March 1996)
- 39. Cass R. Sunstein, The Cost-Benefit State (May 1996)
- 40. William M. Landes and Richard A. Posner, The Economics of Legal Disputes Over the Ownership of Works of Art and Other Collectibles (July 1996)
- 41. John R. Lott, Jr. and David B. Mustard, Crime, Deterrence, and Right-to-Carry Concealed Handguns (August 1996)
- 42. Cass R. Sunstein, Health-Health Tradeoffs (September 1996)
- 43. G. Baird, The Hidden Virtues of Chapter 11: An Overview of the Law and Economics of Financially Distressed Firms (March 1997)
- 44. Richard A. Posner, Community, Wealth, and Equality (March 1997)
- 45. William M. Landes, The Art of Law and Economics: An Autobiographical Essay (March 1997)
- 46. Cass R. Sunstein, Behavioral Analysis of Law (April 1997)
- 47. John R. Lott, Jr. and Kermit Daniel, Term Limits and Electoral Competitiveness: Evidence from California's State Legislative Races (May 1997)
- 48. Randal C. Picker, Simple Games in a Complex World: A Generative Approach to the Adoption of Norms (June 1997)
- Richard A. Epstein, Contracts Small and Contracts Large: Contract Law through the Lens of Laissez-Faire (August 1997)
- 50. Cass R. Sunstein, Daniel Kahneman, and David Schkade, Assessing Punitive Damages (with Notes on Cognition and Valuation in Law) (December 1997)
- 51. William M. Landes, Lawrence Lessig, and Michael E. Solimine, Judicial Influence: A Citation Analysis of Federal Courts of Appeals Judges (January 1998)
- 52. John R. Lott, Jr., A Simple Explanation for Why Campaign Expenditures are Increasing: The Government is Getting Bigger (February 1998)
- 53. Richard A. Posner, Values and Consequences: An Introduction to Economic Analysis of Law (March 1998)
- 54. Denise DiPasquale and Edward L. Glaeser, Incentives and Social Capital: Are Homeowners Better Citizens? (April 1998)
- 55. Christine Jolls, Cass R. Sunstein, and Richard Thaler, A Behavioral Approach to Law and Economics (May 1998)
- 56. John R. Lott, Jr., Does a Helping Hand Put Others At Risk?: Affirmative Action, Police Departments, and Crime (May 1998)
- 57. Cass R. Sunstein and Edna Ullmann-Margalit, Second-Order Decisions (June 1998)
- Jonathan M. Karpoff and John R. Lott, Jr., Punitive Damages: Their Determinants, Effects on Firm Value, and the Impact of Supreme Court and Congressional Attempts to Limit Awards (July 1998)
- 59. Kenneth W. Dam, Self-Help in the Digital Jungle (August 1998)
- 60. John R. Lott, Jr., How Dramatically Did Women's Suffrage Change the Size and Scope of Government? (September 1998)
- 61. Kevin A. Kordana and Eric A. Posner, A Positive Theory of Chapter 11 (October 1998)
- 62. David A. Weisbach, Line Drawing, Doctrine, and Efficiency in the Tax Law (November 1998)
- 63. Jack L. Goldsmith and Eric A. Posner, A Theory of Customary International Law (November 1998)
- 64. John R. Lott, Jr., Public Schooling, Indoctrination, and Totalitarianism (December 1998)
- Cass R. Sunstein, Private Broadcasters and the Public Interest: Notes Toward A "Third Way" (January 1999)
- 66. Richard A. Posner, An Economic Approach to the Law of Evidence (February 1999)
- 67. Yannis Bakos, Erik Brynjolfsson, Douglas Lichtman, Shared Information Goods (February 1999)
- 68. Kenneth W. Dam, Intellectual Property and the Academic Enterprise (February 1999)
- 69. Gertrud M. Fremling and Richard A. Posner, Status Signaling and the Law, with Particular Application to Sexual Harassment (March 1999)
- 70. Cass R. Sunstein, Must Formalism Be Defended Empirically? (March 1999)

- 71. Jonathan M. Karpoff, John R. Lott, Jr., and Graeme Rankine, Environmental Violations, Legal Penalties, and Reputation Costs (March 1999)
- 72. Matthew D. Adler and Eric A. Posner, Rethinking Cost-Benefit Analysis (April 1999)
- 73. John R. Lott, Jr. and William M. Landes, Multiple Victim Public Shooting, Bombings, and Right-to-Carry Concealed Handgun Laws: Contrasting Private and Public Law Enforcement (April 1999)
- Lisa Bernstein, The Questionable Empirical Basis of Article 2's Incorporation Strategy: A Preliminary Study (May 1999)
- 75. Richard A. Epstein, Deconstructing Privacy: and Putting It Back Together Again (May 1999)
- 76. William M. Landes, Winning the Art Lottery: The Economic Returns to the Ganz Collection (May 1999)
- 77. Cass R. Sunstein, David Schkade, and Daniel Kahneman, Do People Want Optimal Deterrence? (June 1999)
- 78. Tomas J. Philipson and Richard A. Posner, The Long-Run Growth in Obesity as a Function of Technological Change (June 1999)
- 79. David A. Weisbach, Ironing Out the Flat Tax (August 1999)
- 80. Eric A. Posner, A Theory of Contract Law under Conditions of Radical Judicial Error (August 1999)
- David Schkade, Cass R. Sunstein, and Daniel Kahneman, Are Juries Less Erratic than Individuals? Deliberation, Polarization, and Punitive Damages (September 1999)
- 82. Cass R. Sunstein, Nondelegation Canons (September 1999)
- 83. Richard A. Posner, The Theory and Practice of Citations Analysis, with Special Reference to Law and Economics (September 1999)
- 84. Randal C. Picker, Regulating Network Industries: A Look at Intel (October 1999)
- 85. Cass R. Sunstein, Cognition and Cost-Benefit Analysis (October 1999)
- 86. Douglas G. Baird and Edward R. Morrison, Optimal Timing and Legal Decisionmaking: The Case of the Liquidation Decision in Bankruptcy (October 1999)
- Gertrud M. Fremling and Richard A. Posner, Market Signaling of Personal Characteristics (November 1999)
- 88. Matthew D. Adler and Eric A. Posner, Implementing Cost-Benefit Analysis When Preferences Are Distorted (November 1999)
- 89. Richard A. Posner, Orwell versus Huxley: Economics, Technology, Privacy, and Satire (November 1999)
- 90. David A. Weisbach, Should the Tax Law Require Current Accrual of Interest on Derivative Financial Instruments? (December 1999)
- 91. Cass R. Sunstein, The Law of Group Polarization (December 1999)
- 92. Eric A. Posner, Agency Models in Law and Economics (January 2000)
- 93. Karen Eggleston, Eric A. Posner, and Richard Zeckhauser, Simplicity and Complexity in Contracts (January 2000)
- 94. Douglas G. Baird and Robert K. Rasmussen, Boyd's Legacy and Blackstone's Ghost (February 2000)
- 95. David Schkade, Cass R. Sunstein, Daniel Kahneman, Deliberating about Dollars: The Severity Shift (February 2000)
- 96. Richard A. Posner and Eric B. Rasmusen, Creating and Enforcing Norms, with Special Reference to Sanctions (March 2000)
- 97. Douglas Lichtman, Property Rights in Emerging Platform Technologies (April 2000)
- 98. Cass R. Sunstein and Edna Ullmann-Margalit, Solidarity in Consumption (May 2000)
- 99. David A. Weisbach, An Economic Analysis of Anti-Tax Avoidance Laws (May 2000, revised May 2002)
- 100. Cass R. Sunstein, Human Behavior and the Law of Work (June 2000)
- 101. William M. Landes and Richard A. Posner, Harmless Error (June 2000)
- 102. Robert H. Frank and Cass R. Sunstein, Cost-Benefit Analysis and Relative Position (August 2000)
- 103. Eric A. Posner, Law and the Emotions (September 2000)
- 104. Cass R. Sunstein, Cost-Benefit Default Principles (October 2000)

- 105. Jack Goldsmith and Alan Sykes, The Dormant Commerce Clause and the Internet (November 2000)
- 106. Richard A. Posner, Antitrust in the New Economy (November 2000)
- 107. Douglas Lichtman, Scott Baker, and Kate Kraus, Strategic Disclosure in the Patent System (November 2000)
- 108. Jack L. Goldsmith and Eric A. Posner, Moral and Legal Rhetoric in International Relations: A Rational Choice Perspective (November 2000)
- 109. William Meadow and Cass R. Sunstein, Statistics, Not Experts (December 2000)
- 110. Saul Levmore, Conjunction and Aggregation (December 2000)
- 111. Saul Levmore, Puzzling Stock Options and Compensation Norms (December 2000)
- 112. Richard A. Epstein and Alan O. Sykes, The Assault on Managed Care: Vicarious Liability, Class Actions and the Patient's Bill of Rights (December 2000)
- 113. William M. Landes, Copyright, Borrowed Images and Appropriation Art: An Economic Approach (December 2000)
- 114. Cass R. Sunstein, Switching the Default Rule (January 2001)
- 115. George G. Triantis, Financial Contract Design in the World of Venture Capital (January 2001)
- 116. Jack Goldsmith, Statutory Foreign Affairs Preemption (February 2001)
- 117. Richard Hynes and Eric A. Posner, The Law and Economics of Consumer Finance (February 2001)
- 118. Cass R. Sunstein, Academic Fads and Fashions (with Special Reference to Law) (March 2001)
- 119. Eric A. Posner, Controlling Agencies with Cost-Benefit Analysis: A Positive Political Theory Perspective (April 2001)
- 120. Douglas G. Baird, Does Bogart Still Get Scale? Rights of Publicity in the Digital Age (April 2001)
- 121. Douglas G. Baird and Robert K. Rasmussen, Control Rights, Priority Rights and the Conceptual Foundations of Corporate Reorganization (April 2001)
- 122. David A. Weisbach, Ten Truths about Tax Shelters (May 2001)
- 123. William M. Landes, What Has the Visual Arts Rights Act of 1990 Accomplished? (May 2001)
- 124. Cass R. Sunstein, Social and Economic Rights? Lessons from South Africa (May 2001)
- 125. Christopher Avery, Christine Jolls, Richard A. Posner, and Alvin E. Roth, The Market for Federal Judicial Law Clerks (June 2001)
- 126. Douglas G. Baird and Edward R. Morrison, Bankruptcy Decision Making (June 2001)
- 127. Cass R. Sunstein, Regulating Risks after ATA (June 2001)
- 128. Cass R. Sunstein, The Laws of Fear (June 2001)
- 129. Richard A. Epstein, In and Out of Public Solution: The Hidden Perils of Property Transfer (July 2001)
- 130. Randal C. Picker, Pursuing a Remedy in *Microsoft*: The Declining Need for Centralized Coordination in a Networked World (July 2001)
- 131. Cass R. Sunstein, Daniel Kahneman, David Schkade, and Ilana Ritov, Predictably Incoherent Judgments (July 2001)
- 132. Eric A. Posner, Courts Should Not Enforce Government Contracts (August 2001)
- 133. Lisa Bernstein, Private Commercial Law in the Cotton Industry: Creating Cooperation through Rules, Norms, and Institutions (August 2001)
- 134. Richard A. Epstein, The Allocation of the Commons: Parking and Stopping on the Commons (August 2001)
- 135. Cass R. Sunstein, The Arithmetic of Arsenic (September 2001)
- 136. Eric A. Posner, Richard Hynes, and Anup Malani, The Political Economy of Property Exemption Laws (September 2001)
- 137. Eric A. Posner and George G. Triantis, Covenants Not to Compete from an Incomplete Contracts Perspective (September 2001)
- 138. Cass R. Sunstein, Probability Neglect: Emotions, Worst Cases, and Law (November 2001)
- 139. Randall S. Kroszner and Philip E. Strahan, Throwing Good Money after Bad? Board Connections and Conflicts in Bank Lending (December 2001)
- 140. Alan O. Sykes, TRIPs, Pharmaceuticals, Developing Countries, and the Doha "Solution" (February 2002)

- 141. Edna Ullmann-Margalit and Cass R. Sunstein, Inequality and Indignation (February 2002)
- 142. Daniel N. Shaviro and David A. Weisbach, The Fifth Circuit Gets It Wrong in *Compaq v. Commissioner* (February 2002) (Published in *Tax Notes*, January 28, 2002)
- 143. Warren F. Schwartz and Alan O. Sykes, The Economic Structure of Renegotiation and Dispute Resolution in the WTO/GATT System (March 2002, *Journal of Legal Studies* 2002)
- 144. Richard A. Epstein, HIPAA on Privacy: Its Unintended and Intended Consequences (March 2002, forthcoming *Cato Journal*, summer 2002)
- 145. David A. Weisbach, Thinking Outside the Little Boxes (March 2002, *Texas Law Review*)
- 146. Eric A. Posner, Economic Analysis of Contract Law after Three Decades: Success or Failure (March 2002)
- 147. Randal C. Picker, Copyright as Entry Policy: The Case of Digital Distribution (April 2002, The Antitrust Bulletin)
- 148. David A. Weisbach, Taxes and Torts in the Redistribution of Income (April 2002, Coase Lecture February 2002)
- 149. Cass R. Sunstein, Beyond the Precautionary Principle (April 2002)
- 150. Robert W. Hahn and Cass R. Sunstein, A New Executive Order for Improving Federal Regulation? Deeper and Wider Cost-Benefit Analysis (April 2002)
- 151. Douglas Lichtman, Copyright as a Rule of Evidence (May 2002, updated January 2003)
- 152. Richard A. Epstein, Steady the Course: Property Rights in Genetic Material (May 2002; revised March 2003)
- 153. Jack Goldsmith and Cass R. Sunstein, Military Tribunals and Legal Culture: What a Difference Sixty Years Makes (June 2002)
- 154. William M. Landes and Richard A. Posner, Indefinitely Renewable Copyright (July 2002)
- 155. Anne Gron and Alan O. Sykes, Terrorism and Insurance Markets: A Role for the Government as Insurer? (July 2002)
- 156. Cass R. Sunstein and Adrian Vermeule, Interpretation and Institutions (July 2002)
- 157. Cass R. Sunstein, The Rights of Animals: A Very Short Primer (August 2002)
- 158. Cass R. Sunstein, Avoiding Absurdity? A New Canon in Regulatory Law (with Notes on Interpretive Theory) (August 2002)
- 159. Randal C. Picker, From Edison to the Broadcast Flag: Mechanisms of Consent and Refusal and the Propertization of Copyright (September 2002)
- 160. Eric A. Posner, A Theory of the Laws of War (September 2002)
- 161 Eric A. Posner, Probability Errors: Some Positive and Normative Implications for Tort and Contract Law (September 2002)
- 162. Lior Jacob Strahilevitz, Charismatic Code, Social Norms, and the Emergence of Cooperation on the File-Swapping Networks (September 2002)
- 163. David A. Weisbach, Does the X-Tax Mark the Spot? (September 2002)
- 164. Cass R. Sunstein, Conformity and Dissent (September 2002)
- 165. Cass R. Sunstein, Hazardous Heuristics (October 2002)
- 166. Douglas Lichtman, Uncertainty and the Standard for Preliminary Relief (October 2002)
- 167. Edward T. Swaine, Rational Custom (November 2002)
- 168. Julie Roin, Truth in Government: Beyond the Tax Expenditure Budget (November 2002)
- 169. Avraham D. Tabbach, Criminal Behavior: Sanctions and Income Taxation: An Economic Analysis (November 2002)
- 170. Richard A. Epstein, In Defense of "Old" Public Health: The Legal Framework for the Regulation of Public Health (December 2002)
- 171. Richard A. Epstein, Animals as Objects, or Subjects, of Rights (December 2002)
- 172. David A. Weisbach, Taxation and Risk-Taking with Multiple Tax Rates (December 2002)
- 173. Douglas G. Baird and Robert K. Rasmussen, The End of Bankruptcy (December 2002)
- 174. Richard A. Epstein, Into the Frying Pan: Standing and Privity under the Telecommunications Act of 1996 and Beyond (December 2002)
- 175. Douglas G. Baird, In Coase's Footsteps (January 2003)

- 176. David A. Weisbach, Measurement and Tax Depreciation Policy: The Case of Short-Term Assets (January 2003)
- 177. Randal C. Picker, Understanding Statutory Bundles: Does the Sherman Act Come with the 1996 Telecommunications Act? (January 2003)
- 178. Douglas Lichtman and Randal C. Picker, Entry Policy in Local Telecommunications: *Iowa Utilities* and *Verizon* (January 2003)
- 179. William Landes and Douglas Lichtman, Indirect Liability for Copyright Infringement: An Economic Perspective (February 2003)
- 180. Cass R. Sunstein, Moral Heuristics (March 2003)
- 181. Amitai Aviram, Regulation by Networks (March 2003)
- 182. Richard A. Epstein, Class Actions: Aggregation, Amplification *and* Distortion (April 2003)
- 183. Richard A. Epstein, The "Necessary" History of Property and Liberty (April 2003)
- 184. Eric A. Posner, Transfer Regulations and Cost-Effectiveness Analysis (April 2003)
- 185. Cass R. Sunstein and Richard H. Thaler, Libertarian Paternalizm Is Not an Oxymoron (May 2003)
- Alan O. Sykes, The Economics of WTO Rules on Subsidies and Countervailing Measures (May 2003)
- 187. Alan O. Sykes, The Safeguards Mess: A Critique of WTO Jurisprudence (May 2003)
- 188. Alan O. Sykes, International Trade and Human Rights: An Economic Perspective (May 2003)
- 189. Saul Levmore and Kyle Logue, Insuring against Terrorism—and Crime (June 2003)
- 190. Richard A. Epstein, Trade Secrets as Private Property: Their Constitutional Protection (June 2003)
- 191. Cass R. Sunstein, Lives, Life-Years, and Willingness to Pay (June 2003)
- 192. Amitai Aviram, The Paradox of Spontaneous Formation of Private Legal Systems (July 2003)
- 193. Robert Cooter and Ariel Porat, Decreasing Liability Contracts (July 2003)
- 194. David A. Weisbach and Jacob Nussim, The Integration of Tax and Spending Programs (September 2003)
- 195. William L. Meadow, Anthony Bell, and Cass R. Sunstein, Statistics, Not Memories: What Was the Standard of Care for Administering Antenatal Steroids to Women in Preterm Labor between 1985 and 2000? (September 2003)
- 196. Cass R. Sunstein, What Did *Lawrence* Hold? Of Autonomy, Desuetude, Sexuality, and Marriage (September 2003)
- Randal C. Picker, The Digital Video Recorder: Unbundling Advertising and Content (September 2003)
- 198. Cass R. Sunstein, David Schkade, and Lisa Michelle Ellman, Ideological Voting on Federal Courts of Appeals: A Preliminary Investigation (September 2003)
- 199. Avraham D. Tabbach, The Effects of Taxation on Income Producing Crimes with Variable Leisure Time (October 2003)
- 200. Douglas Lichtman, Rethinking Prosecution History Estoppel (October 2003)
- 201. Douglas G. Baird and Robert K. Rasmussen, Chapter 11 at Twilight (October 2003)
- 202. David A. Weisbach, Corporate Tax Avoidance (January 2004)
- 203. David A. Weisbach, The (Non)Taxation of Risk (January 2004)
- 204. Richard A. Epstein, Liberty versus Property? Cracks in the Foundations of Copyright Law (April 2004)
- 205. Lior Jacob Strahilevitz, The Right to Destroy (January 2004)
- 206. Eric A. Posner and John C. Yoo, A Theory of International Adjudication (February 2004)
- 207. Cass R. Sunstein, Are Poor People Worth Less Than Rich People? Disaggregating the Value of Statistical Lives (February 2004)
- 208. Richard A. Epstein, Disparities and Discrimination in Health Care Coverage; A Critique of the Institute of Medicine Study (March 2004)
- 209. Richard A. Epstein and Bruce N. Kuhlik, Navigating the Anticommons for Pharmaceutical Patents: Steady the Course on Hatch-Waxman (March 2004)
- 210. Richard A. Esptein, The Optimal Complexity of Legal Rules (April 2004)
- 211. Eric A. Posner and Alan O. Sykes, Optimal War and Jus Ad Bellum (April 2004)
- 212. Alan O. Sykes, The Persistent Puzzles of Safeguards: Lessons from the Steel Dispute (May 2004)

- 213. Luis Garicano and Thomas N. Hubbard, Specialization, Firms, and Markets: The Division of Labor within and between Law Firms (April 2004)
- 214. Luis Garicano and Thomas N. Hubbard, Hierarchies, Specialization, and the Utilization of Knowledge: Theory and Evidence from the Legal Services Industry (April 2004)
- 215. James C. Spindler, Conflict or Credibility: Analyst Conflicts of Interest and the Market for Underwriting Business (July 2004)
- 216. Alan O. Sykes, The Economics of Public International Law (July 2004)
- 217. Douglas Lichtman and Eric Posner, Holding Internet Service Providers Accountable (July 2004)
- 218. Shlomo Benartzi, Richard H. Thaler, Stephen P. Utkus, and Cass R. Sunstein, Company Stock, Market Rationality, and Legal Reform (July 2004)
- 219. Cass R. Sunstein, Group Judgments: Deliberation, Statistical Means, and Information Markets (August 2004, revised October 2004)
- 220. Cass R. Sunstein, Precautions against What? The Availability Heuristic and Cross-Cultural Risk Perceptions (August 2004)
- 221. M. Todd Henderson and James C. Spindler, Corporate Heroin: A Defense of Perks (August 2004)
- 222. Eric A. Posner and Cass R. Sunstein, Dollars and Death (August 2004)
- 223. Randal C. Picker, Cyber Security: Of Heterogeneity and Autarky (August 2004)
- 224. Randal C. Picker, Unbundling Scope-of-Permission Goods: When Should We Invest in Reducing Entry Barriers? (September 2004)
- 225. Christine Jolls and Cass R. Sunstein, Debiasing through Law (September 2004)
- 226. Richard A. Posner, An Economic Analysis of the Use of Citations in the Law (2000)
- 227. Cass R. Sunstein, Cost-Benefit Analysis and the Environment (October 2004)
- 228. Kenneth W. Dam, Cordell Hull, the Reciprocal Trade Agreement Act, and the WTO (October 2004)
- 229. Richard A. Posner, The Law and Economics of Contract Interpretation (November 2004)
- 230. Lior Jacob Strahilevitz, A Social Networks Theory of Privacy (December 2004)
- 231. Cass R. Sunstein, Minimalism at War (December 2004)
- 232. Douglas Lichtman, How the Law Responds to Self-Help (December 2004)
- 233. Eric A. Posner, The Decline of the International Court of Justice (December 2004)
- 234. Eric A. Posner, Is the International Court of Justice Biased? (December 2004)
- 235. Alan O. Sykes, Public vs. Private Enforcement of International Economic Law: Of Standing and Remedy (February 2005)
- 236. Douglas G. Baird and Edward R. Morrison, Serial Entrepreneurs and Small Business Bankruptcies (March 2005)
- 237. Eric A. Posner, There Are No Penalty Default Rules in Contract Law (March 2005)
- Randal C. Picker, Copyright and the DMCA: Market Locks and Technological Contracts (March 2005)
- 239. Cass R. Sunstein and Adrian Vermeule, Is Capital Punishment Morally Required? The Relevance of Life-Life Tradeoffs (March 2005)
- 240. Alan O. Sykes, Trade Remedy Laws (March 2005)
- 241. Randal C. Picker, Rewinding *Sony*: The Evolving Product, Phoning Home, and the Duty of Ongoing Design (March 2005)
- 242. Cass R. Sunstein, Irreversible and Catastrophic (April 2005)
- 243. James C. Spindler, IPO Liability and Entrepreneurial Response (May 2005)
- 244. Douglas Lichtman, Substitutes for the Doctrine of Equivalents: A Response to Meurer and Nard (May 2005)
- 245. Cass R. Sunstein, A New Progressivism (May 2005)
- 246. Douglas G. Baird, Property, Natural Monopoly, and the Uneasy Legacy of *INS v. AP* (May 2005)
- 247. Douglas G. Baird and Robert K. Rasmussen, Private Debt and the Missing Lever of Corporate Governance (May 2005)
- 248. Cass R. Sunstein, Administrative Law Goes to War (May 2005)
- 249. Cass R. Sunstein, Chevron Step Zero (May 2005)
- 250. Lior Jacob Strahilevitz, Exclusionary Amenities in Residential Communities (July 2005)

- 251. Joseph Bankman and David A. Weisbach, The Superiority of an Ideal Consumption Tax over an Ideal Income Tax (July 2005)
- 252. Cass R. Sunstein and Arden Rowell, On Discounting Regulatory Benefits: Risk, Money, and Ingergenerational Equity (July 2005)
- 253. Cass R. Sunstein, Boundedly Rational Borrowing: A Consumer's Guide (July 2005)
- 254. Cass R. Sunstein, Ranking Law Schools: A Market Test? (July 2005)
- 255. Dexter Samida and David A. Weisbach, Paretian Intergenerational Discounting (August 2005)